

Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL089348

Key Points:

- Since 2000, the March-April-May sea surface temperature variability has weakened by more than 30% in the Angola-Benguela area
- Since 2000, the equatorial remote forcing has less influence on Angola-Benguela sea surface temperatures
- Since 2000, the Angola-Benguela local forcing is likely to impact the surface layer more due to a shallower stratification maximum

Supporting Information:

Supporting Information S1

Correspondence to:

A. Prigent, aprigent@geomar.de

Citation:

Prigent, A., Imbol Koungue, R. A., Lübbecke, J. F., Brandt, P., & Latif, M. (2020). Origin of weakened interannual sea surface temperature variability in the southeastern tropical Atlantic Ocean. *Geophysical Research Letters*, 47, e2020GL089348. https://doi.org/ 10.1029/2020GL089348

Received 16 JUN 2020 Accepted 30 SEP 2020 Accepted article online 7 OCT 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

PRIGENT ET AL.

Origin of Weakened Interannual Sea Surface Temperature Variability in the Southeastern Tropical Atlantic Ocean

Arthur Prigent¹, Rodrigue Anicet Imbol Koungue¹, Joke F. Lübbecke^{1,2}, Peter Brandt^{1,2}, and Mojib Latif^{1,2}

¹GEOMAR Helmholtz Centre for Ocean Research Kiel, Kiel, Germany, ²Faculty of Mathematics and Natural Sciences, Christian Albrechts University, Kiel, Germany

Abstract Observations and reanalysis products are used to investigate the substantial weakening in the southeastern tropical Atlantic sea surface temperature (SST) variability since 2000. Relative to 1982–1999, the March-April-May SST variability in the Angola-Benguela area (ABA) has decreased by more than 30%. Both equatorial remote forcing and local forcing are known to play an important role in driving SST variability in the ABA. Compared to 1982–1999, since 2000, equatorial remote forcing had less influence on ABA SSTs, whereas local forcing has become more important. In particular, the robust correlation that existed between the equatorial zonal wind stress and the ABA SSTs has substantially weakened, suggesting less influence of Kelvin waves on ABA SSTs. Moreover, the strong correlation linking the South Atlantic Anticyclone and the ABA SSTs has reduced. Finally, multidecadal surface warming of the ABA could also have played a role in the weakening of the interannual SST variability.

Plain Language Summary Every few years, the southeastern tropical Atlantic Ocean experiences anomalous sea temperatures that affect fisheries and rainfall. Using observations and reanalysis data, we quantify the Angola-Benguela area (ABA) sea surface temperature (SST) variability during the last decades. Relative to 1982–1999, the March-April-May SST variability in the ABA has decreased by more than 30% since 2000. Remote equatorial forcing through equatorially and coastally trapped oceanic waves and variations in the local winds are the main drivers of ABA SST variability. Since 2000, we find that ABA SSTs are less connected to equatorial winds and exhibit a weaker link with the South Atlantic Anticyclone. Finally, the surface warming of the ABA observed during the post-2000 period also could have played a role in the weakening interannual SST variability.

1. Introduction

The tropical Atlantic Ocean sea surface temperatures (SSTs) have warmed substantially over the last decades (Servain et al., 2014; Tokinaga & Xie, 2011; Vizy & Cook, 2016) and particularly in the southeastern tropical Atlantic. In association with the basin-wide warming, which is most pronounced in boreal summer reducing the annual cycle through a positive ocean-atmosphere feedback, a strong reduction of the equatorial Atlantic SST variability was reported by Tokinaga and Xie (2011) over the period 1950–2009. However, multidecadal variability is large. Nnamchi et al. (2020) found that SSTs during the satellite era 1979–2018 exhibit a warming hole over the equatorial Atlantic cold tongue region in boreal summer. This lack of surface warming denotes an 11% amplification of the mean SST annual cycle in that region. Recently, Prigent et al. (2020), using observations and reanalysis products, found that the equatorial Atlantic interannual SST variability in May-June-July has reduced by 31% during 2000–2017 relative to 1982–1999. Interannual SST variability in the equatorial Atlantic and Angola-Benguela area (ABA) are strongly connected (Hu & Huang, 2007; Lübbecke et al., 2010; Reason et al., 2006). Hence, it may well be that the SST variability in the southeastern tropical Atlantic also has undergone a reduction since 2000, which is the topic of this study.

SSTs along the coasts of Namibia and Angola are characterized by a strong seasonal cycle, with warmest SSTs in March-April-May (MAM), modulated by variability from subseasonal to decadal time scales (Bachèlery et al., 2019). The main features of interannual SST variability over the ABA (8°E to the coast; 10–20°S, blue box in Figure 1a) are warm and cold events, the so-called Benguela Niños and Benguela Niñas (Shannon et al., 1986), respectively. Anomalous surface temperatures along the coasts of Angola and Namibia, typically lasting for a few months and peaking in MAM, impact the regional climate (Hansingo & Reason, 2009;





Figure 1. (a) Difference of MAM SST anomalies (SSTa) standard deviation between 2000–2017 and 1982–1999 from ERA5. Dots indicate where the standard deviations between the two periods are significantly different at the 95% level according to a Welch's *t* test. (b) Time series of monthly ABA-averaged ERA5 SSTa. (c) Ensemble mean standard deviation of ABA-averaged SSTa as function of the calendar month during 1982–1999 (red) and 2000–2017 (blue). The six SST products used are indicated in Table S1. (d) Autocorrelation function of the monthly ABA SSTa during 1982–1999 (red) and 2000–2017 (blue).

Koseki & Imbol Koungue, 2020; Lutz et al., 2013; Rouault et al., 2003, 2007) as well as marine ecosystems and fisheries (Bachèlery et al., 2016, 2019; Binet et al., 2001; Gammelsrød et al., 1998).

Such events are mainly driven by two mechanisms: (1) remote forcing from the equatorial Atlantic and (2) local atmospheric forcing. The remote forcing is associated with the fluctuations of the trade winds over the western and central parts of the equatorial Atlantic, triggering eastward propagating equatorial Kelvin waves (EKWs). When reaching the West African coast, part of the energy of the EKWs is transmitted poleward along the West African coast (Polo et al., 2008) as coastally trapped waves (CTWs; Clarke, 1983), which affect local stratification, SST, alongshore currents, and biogeochemical conditions (Bachèlery et al., 2015, 2016, 2019; Illig & Bachèlery, 2019; Illig, Bachèlery, & Cadier, 2018; Illig, Cadier, et al., 2018; Rouault, 2012; Rouault et al., 2018). Bachèlery et al. (2019) showed in an ocean modeling study that remote forcing from the equatorial Atlantic explains around 50% (70%) of the interannual SST (sea level anomaly) variability between 10°S and 20°S along the West African coast. Richter et al. (2010) showed, using observations and a coupled general circulation model, that Benguela Niños also can be forced by weakened local alongshore winds related to the strength of the South Atlantic Anticyclone (SAA). Moreover, Lübbecke et al. (2019) have demonstrated that the warm event of 2016 off Angola and Namibia was generated by a combination of local processes, that is, reduction of alongshore winds and local upwelling, anomalous heat fluxes, freshwater input, and meridional advection.

Predicting SSTs in the ABA is challenging. Two predictors have been suggested: (1) an index based on SAA strength (Lübbecke et al., 2010) and (2) an index based on EKW activity (Imbol Koungue et al., 2017). Both approaches would allow to predict Benguela Niño and Benguela Niña events at 1 to 2 months lead time.



In this study, we document a marked reduction of the interannual SST variability in the ABA since 2000. Possible links to remote forcing from the equatorial Atlantic and to the SAA and implications for the predictability of Benguela Niños and Benguela Niñas are discussed.

2. Data and Methods

2.1. Data

2.1.1. Ocean Reanalyses

Three ocean reanalysis SST products with monthly resolution are used: (1) Climate Forecast System Reanalysis Versions 1 and 2 (CFSR; Saha et al., 2014) available at 0.5° horizontal resolution and spanning the period from January 1979 to August 2019. (2) European Centre for Medium-range Weather Forecast (ECMWF) Re-Analysis 5 (ERA5; Hersbach et al., 2020) available at 0.25° horizontal resolution for the period January 1979 to January 2020. (3) Ocean Reanalysis System Version 4 (ORA-S4; Balmaseda et al., 2013) from the ECMWF available at 1° horizontal resolution and spanning the period January 1958 to December 2017. Wind stress and sea level pressure (SLP) are taken from ERA5. The 20°C isotherm depth (z_{20}) is taken from ORA-S4.

2.1.2. Observations

Observational SST products used are (1) Hadley Centre Sea Ice and SST data set Version 1.1 (HadI-SST; Rayner et al., 2003) available at 1° horizontal resolution for the period January 1870 to December 2019. (2) Centennial in situ Observation-Based Estimates SST (COBE-SST; Ishii et al., 2005) available at 1° horizontal resolution and spanning the period January 1891 to December 2019. (3) Optimum Interpolation SST Version 2 (OI-SST; Reynolds et al., 2007) available at 0.25° horizontal resolution for the period September 1981 to December 2019. Table S1 in the supporting information provides an overview of the data sets and variables used.

2.2. Methods

In order to investigate the changes in the interannual SST variability in the ABA since 2000, we compare the periods January 1982 to December 1999 and January 2000 to December 2017. The year 2000 was chosen as the year of separation because major shifts occurred around this year both in the tropical Atlantic (Prigent et al., 2020) and tropical Pacific (Hu et al., 2013, 2017, 2020; Li et al., 2019; Lübbecke & McPhaden, 2014; McPhaden, 2012). All analyses use monthly-mean anomalies computed by subtracting the climatological monthly-mean seasonal cycle derived separately for each data set and period. Prior to all analyses, linear trends calculated over the entire period 1982–2017 were removed.

3. SST Variability

The magnitude of the interannual SST variability in the southeastern tropical Atlantic, as measured by the standard deviation, has undergone a strong multidecadal reduction (Figure 1a). The time series of the SST anomalies (SSTa) averaged over the ABA (Figure 1b) exhibits a clear change in character from the pre-2000 period to the post-2000 period. The seasonal cycle of the standard deviation of the SSTa in the ABA exhibits a distinct maximum in MAM (Figure 1c), when most of the Benguela Niños/Niñas occur (Imbol Koungue et al., 2017, 2019). However, the standard deviations during 2000–2017 are much smaller. There is still considerable interannual variability after 2000, but major events are fewer and SST fluctuations less persistent (Figure 1d).

While the MAM SST in the ABA has warmed by 0.3 K from 1982–1999 to 2000–2017 (Figure S1a), the meridional gradient of MAM SST in the ABA remained nearly constant (Figure S1b). Vizy et al. (2018) showed a poleward trend of the Angola-Benguela Frontal Zone (ABFZ), which could have influenced the ABA SST variability as the area of variability would shift with it. However, the observed reduction in SST variability occurred over a relatively large region. While the reduction is significant mainly to the north of the ABFZ, it extends almost along the whole Angolan and Namibian coasts (Figure 1a). Therefore, the reduction is not likely associated with a poleward displacement of the ABFZ.

The standard deviation of SSTa in the ABA calculated over all calendar months has weakened by 22.3%, from 0.74 \pm 0.05 K in 1982–1999 to 0.58 \pm 0.07 K in 2000–2017. Strongest reduction occurred in MAM



Table 1

Standard Deviation	of the ABA-Averaged	d SSTa (K)	for the Periods	1982–1999 and	2000-2017
Dianaana Devianon	of the month invertiget	a DD1 a (11)	joi inc i crious	1)02 1))) unu	2000 2017

	1982–1999		2000-2017		Reduction (%)	
Product	All month	MAM	All month	MAM	All month	MAM
CFSR	0.75	1.15	0.64	0.83	15	28
ERA5	0.75	1.10	0.56	0.80	25	27
HadI-SST	0.72	0.91	0.48	0.63	33	31
COBE-SST	0.64	0.90	0.48	0.58	25	36
OI-SST	0.78	1.23	0.66	0.84	16	32
ORA-S4	0.78	1.18	0.63	0.84	20	29
EM	0.74 ± 0.05	1.08 ± 0.13	0.58 ± 0.07	0.75 ± 0.11	22.3 ± 6.2	30.5 ± 3

Note. EM denotes the ensemble mean of all products (Table S1).



Figure 2. The vectors represent the regressions of detrended ERA5 wind stress anomalies on ERA5 ABA-averaged (8°E to the coast, $20-10^{\circ}$ S; blue box) SSTa with winds leading by 1 month for (a) 1982–1999 and (b) 2000–2017. Regressions have been calculated for each wind stress component separately. The color shading depicts the magnitude of the vectors. Black (gray) arrows indicate pointwise significant (not significant) regressions for both components at the 95% level according to a Student's *t* test. (c, d) Regression coefficients of detrended ERA5 wind stress curl anomalies on ERA5 ABA-averaged SSTa with the wind stress curl anomalies leading by 1 month during 1982–1999 and 2000–2017, respectively. Displayed regressions are significant at the 95% level according to the Student's *t* test.





Figure 3. (a) Time series of monthly ATL3 ORA-S4 20°C isotherm depth anomalies (z_{20a}). (b) Cross correlation of monthly ATL3 z_{20a} and ABA SSTa for 1982–1999 (red) and 2000–2017 (blue). Dots indicate correlations that are significant at the 95% level. (c, d) Regression coefficients of the detrended ORA-S4 SSTa on ATL3 ORA-S4 z_{20a} with the z_{20a} leading by 1 month during (c) 1982–1999 and (d) 2000–2017. Only statistically significant regressions at the 95% level are shown. Statistical significance is assessed by the Student's *t* test.

(Table 1). The MAM standard deviation was 1.08 ± 0.13 K during 1982–1999 and decreased by 30.5% to 0.75 ± 0.11 K during 2000–2017 (Table 1).

4. Relative Roles of Equatorial and Local Forcing

During 1982–1999, the ABA SSTa are mainly linked to zonal wind stress fluctuations in the western equatorial Atlantic, as shown by linear regression on ABA SSTa (Figures 2a and S2a). Equatorial zonal wind stress fluctuations can generate EKWs that propagate eastward along the equator. At the West African coast, part of the EKW energy is transmitted poleward as CTWs which can trigger coastal warm/cold events (Bachèlery et al., 2019; Florenchie et al., 2003, 2004; Illig et al., 2004; Imbol Koungue et al., 2017, 2019; Lübbecke et al., 2010). In contrast, during 2000–2017 (Figures 2b and S2b), the link between western equatorial zonal wind stress fluctuations and coastal SSTa is considerably weaker, which is consistent with the reduced equatorial wind stress variability described in Prigent et al. (2020). Instead, relatively large regression coefficients are found in the southeastern tropical Atlantic close to the West African coast (Figure 2b), suggesting that the role of local meridional wind stress fluctuations in driving interannual SST variability in the ABA has increased. This is further supported by the strengthened link between ABA SSTa and near-coastal wind stress curl anomalies along the Angolan and Namibian coasts when comparing 1982–1999 (Figure 2c) to 2000–2017 (Figure 2d). We also observe a magnitude maximum of the regression coefficients around 20°S and 20°W (Figure 2b), which, however, is not statistically significant at 95% according to a Student's *t* test.

We next examine the impact of the equatorial zonal wind stress on thermocline depth. Variability in the latter serves here to assess the role of EKW activity (Imbol Koungue et al., 2017).





Figure 4. Time series of interannual anomalies of February SAA index (SLP (hPa) averaged over 40°S to 20°S and 30°W to 10°W; inverted, full line) and March ABA SST (°C, dashed line) from ERA5 during 1982–1999 (red) and 2000–2017 (blue).

The standard deviation of the thermocline depth anomalies (z_{20a}) in the ATL3 box (20°W to 0°, 3°N to 3°S; Figure 3a) has reduced by 15.4% from 5.19 m during 1982–1999 to 4.39 m during 2000–2017. Further, the relationship between ATL3 z_{20a} and ABA SSTa also has markedly weakened after 2000 (Figure 3b). In fact, during 1982–1999, the ATL3 z_{20a} was leading ABA SSTa by 1 month with a correlation of 0.52 whereas the correlation at the same lead time dropped to 0.28 during 2000–2017. These results suggest that after 2000 the importance of remote equatorial forcing of ABA SSTa by EKWs has substantially reduced. Consistent with this, the major warm event of 2016 resulted from a combination of different local processes (Lübbecke et al., 2019). The results shown in Figure 3b also imply that an index based on equatorial Atlantic oceanic variability has become a less skillful predictor for the ABA SST variability after 2000: While the shape of the cross-correlation function remained unchanged, the magnitude of the correlations has dropped considerably.

The link between the ABA SSTa and the equatorial ocean dynamics can be estimated by regressing the SSTa on the (detrended) ATL3 z_{20a} taken 1 month earlier. This link has weakened at the equator from 1982–1999 (Figure 3c) to 2000–2017 (Figure 3d), especially between 20°W and 5°E and along the West African coast between 5°S and 20°S. This again suggests that after 2000, equatorial thermocline displacement anomalies have less impact on the ABA SSTs. However, reanalysis products are known to exhibit large biases in the tropical Atlantic region (Huang et al., 2007; Kumar & Hu, 2012; Tchipalanga et al., 2018). High-resolution ocean models forced by the history of observed wind stress could be a way out of this dilemma.

Lübbecke et al. (2010) showed that western equatorial Atlantic zonal wind stress variations are linked to the variations of the strength of the SAA. Later, Lübbecke et al. (2014) demonstrated that this link is facilitated through the wind power. Richter et al. (2010) also highlighted the importance of the SAA on the development of warm events off the Angolan and Namibian coasts.

In agreement with Lübbecke et al. (2010), the fluctuations of the strength of the SAA in austral summer during 1982–1999 have been well anticorrelated with the subsequent austral fall ABA SSTs with a correlation coefficient of -0.74 (Figure 4). However, over the period 2000–2017, the correlation coefficient dropped to -0.41. No major shift in the SAA position was found that could explain the diminishing correlation between March ABA SSTa and February SLP anomalies (SLPa) from 1982–1999 (Figure S3a) to 2000–2017 (Figure S3b). Moreover, since 2000 a weaker relationship between the SLP-anomaly field and ABA SSTa is evident in the regression maps (Figures S3c and S3d). Hence, the ABA SSTs during 2000–2017 are less influenced by the variations in the strength of the SAA than during 1982–1999. The reason for this weaker relationship is unclear and beyond the scope of this paper.

5. Discussion and Conclusions

This study documents a multidecadal reduction in the interannual SST variability in the ABA during 2000–2017 relative to 1982–1999. The interannual SST variability in the ABA has reduced in the annual mean by 22.3% during 2000–2017 relative to 1982–1999, with the strongest reduction amounting to 30.5% in MAM. The reduced interannual SST variability in the ABA goes along with a smaller influence of remote forcing by equatorial wind stress variability. The zonal wind stress variability over the western equatorial

Atlantic diminished during the recent decades, as reported by Prigent et al. (2020). Lower zonal wind stress variability tends to reduce EKW activity that is an important driver of SST in the ABA. This by itself enhances the relative importance of the local atmospheric forcing through near-coastal meridional wind stress and wind stress curl, if the local factors remain unchanged. However, the quantification of the relative importance of the local and remote wind stress forcing over the last decades remains a challenge due to limited data. Ocean modeling would be an alternative and additionally allow disentangling the contribution of each equatorial baroclinic mode to ABA SST variability.

Another factor that could have contributed to the reduced interannual SST variability in the ABA is the multidecadal surface warming in the region. In MAM, the season of the largest interannual SST variability, the SST in the ABA has warmed by 0.3 K when comparing the pre-2000 period to the post-2000 period. This is consistent with Vizy and Cook (2016) who investigated multidecadal changes in the southeastern tropical Atlantic over the period 1982–2013 and found significant SST warming trends along the Angolan/Namibian coasts of 0.5–1.5 K per 32 years. This multidecadal surface warming was associated with an increased net heat flux from the atmosphere and reduced coastal upwelling. The surface warming and concurrent subsurface cooling (Figures S4a and S4b) might have reduced surface-subsurface coupling in the post-2000 period. There is a shoaling of the stratification maximum in the ABA since 2000 (Figure S4b), which would support a larger sensitivity of the mixed layer to local forcing. Due to large biases in reanalysis products and limited observations (Huang et al., 2007; Kumar & Hu, 2012; Tchipalanga et al., 2018), large uncertainties remain regarding the potential influence of multidecadal changes in stratification on the interannual SST variability in the ABA.

Although some major remotely forced Benguela Niño/Niña events occurred after 2000, Tables S1 and S2 from Imbol Koungue et al. (2019) indicate a smaller number relative to previous decades. Over the period 1982–2015, out of eight (six) warm (cold) events only two (one) took place after 2000. However, over the same time period, out of 15 (7) moderate warm (cold) events 7 (5) occurred between 2000 and 2015. This illustrates that relative to 1982–1999, since 2000 fewer remotely forced major and more moderate events occurred in the ABA.

Numerous other potential factors could have contributed to the reduced SST variability in the ABA. One of them might be the interdecadal shift of El Niño/Southern Oscillation (ENSO) that occurred around 2000 (Hu et al., 2013, 2017, 2020; Lübbecke & McPhaden, 2014; McPhaden, 2012). This shift featured a decrease in ENSO variability, an increase in ENSO frequency (Hu et al., 2013, 2020) as well as a profound westward shift of the Walker Circulation in the equatorial Pacific (Li et al., 2019). The latter might have altered ENSO teleconnections, and thus, could have contributed to the weakened SST variability in the tropical Atlantic. However, the influence of the multidecadal shift in ENSO on the tropical Atlantic SST requires further analyses.

Finally, our results may have implications for the predictability of Benguela Niño/Niña events. In particular, the strong link between equatorial thermocline displacements and ABA SSTa during 1982–1999 has significantly weakened since 2000. In addition, we show that the strong link between the fluctuations in SAA strength and ABA SSTa observed during the pre-2000 period (Lübbecke et al., 2010) has diminished considerably during the post-2000 period. This raises the question as to whether predictors based on equatorial variables or the SAA strength are still useful to forecast Benguela Niño/Niña events.

Data Availability Statement

The surface and ocean profile data sets were taken from the following publicly available sources: OI-SST (https://psl.noaa.gov/data/gridded/data.noaa.oisst.v2.highres.html), ORA-S4 (http://icdc.cen.uni-hamburg.de/ projekte/easy-init/easy-init-ocean.html), ERA5 (https://www.ecmwf.int/en/forecasts/datasets/reanalysis-datasets/east), HadI-SST (https://www.metoffice.gov.uk/hadobs/hadisst/), COBE-SST (https://psl.noaa.gov/ data/gridded/data.cobe2.html), and CFSR (https://climatedataguide.ucar.edu/climate-data/climate-fore-cast-system-reanalysis-cfsr).

References

Bachèlery, M.-L., Illig, S., & Dadou, I. (2015). Interannual variability in the South-East Atlantic Ocean, focusing on the Benguela Upwelling System: Remote versus local forcing. *Journal of Geophysical Research: Oceans*, 121, 284–310. https://doi.org/10.1002/2015JC011168

Acknowledgments

This study was supported by the German Federal Ministry of Education and Research as part of the BANINO project (03F0795A). This paper was also supported by the TRIATLAS project, which has received funding from the European Union's Horizon 2020 research and innovation program under grant agreement No. 817578.

- Bachèlery, M.-L., Illig, S., & Dadou, I. (2016). Forcings of nutrient, oxygen, and primary production interannual variability in the southeast Atlantic Ocean. *Geophysical Research Letters*, 43, 8617–8625. https://doi.org/10.1002/2016GL070288
- Bachèlery, M.-L., Illig, S., & Rouault, M. (2019). Interannual coastal trapped waves in the Angola-Benguela upwelling system and Benguela Niño and Niña events. *Journal of Marine Systems*, 203, 103262. https://doi.org/10.1016/j.jmarsys.2019.103262
- Balmaseda, M. A., Mogensen, K., & Weaver, A. T. (2013). Evaluation of the ECMWF ocean reanalysis system ORAS4. Quarterly Journal of the Royal Meteorological Society, 139, 1132–1161. https://doi.org/10.1002/qj.2063
- Binet, D., Gobert, B., & Maloueki, L. (2001). El Niño-like warm events in the Eastern Atlantic (6°N, 20°S) and fish availability from Congo to Angola (1964–1999). Aquatic Living Resources, 14, 99–113. https://doi.org/10.1016/S0990-7440(01)01105-6
- Clarke, A. J. (1983). The reflection of equatorial waves from oceanic boundaries. *Journal of Physical Oceanography*, 13(7), 1193–1207. https://doi.org/10.1175/1520-0485(1983)013<1193:TROEWF>2.0.CO;2
- Florenchie, P., Lutjeharms, J. R., Reason, C. J. C., Masson, S., & Rouault, M. (2003). The source of Benguela Niños in the South Atlantic Ocean. Geophysical Research Letters, 30(10), 1505. https://doi.org/10.1029/2003GL017172
- Florenchie, P., Reason, C. J. C., Lutjeharms, J. R. E., Rouault, M., Roy, C., & Masson, S. (2004). Evolution of interannual warm and cold events in the southeast Atlantic Ocean. *Journal of Climate*, 17(12), 2318–2334. https://doi.org/10.1175/1520-0442(2004)017<2318: EOIWAC>2.0.CO:2
- Gammelsrød, T., Bartholomae, C. H., Boyer, D. C., Filipe, V. L. L., & O'Toole, M. J. (1998). Intrusion of warm surface layer along the Angolan–Namibian coast in February–March: The 1995 Benguela Niño. South African Journal of Marine Science, 19, 41–56. https://doi. org/10.2989/025776198784126719
- Hansingo, K., & Reason, C. J. C. (2009). Modelling the atmospheric response over southern Africa to SST forcing in the southeast tropical Atlantic and southwest subtropical Indian Oceans. *International Journal of Climatology*, 29, 1001–1012. https://doi.org/10.1002/joc.1919 Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. *Quarterly*

Journal of the Royal Meteorological Society, 146, 1999–2049. https://doi.org/10.1002/qj.3803

- Hu, Z., Kumar, A., Huang, B., Zhu, J., L'Heureux, M., McPhaden, M. J., & Yu, J. (2020). The interdecadal shift of ENSO properties in 1999/ 2000: A review. Journal of Climate, 33, 4441–4462. https://doi.org/10.1175/JCLI-D-19-0316.1
- Hu, Z.-Z., & Huang, B. (2007). Physical processes associated with the tropical Atlantic SST gradient during the anomalous evolution in the southeastern ocean. Journal of Climate, 20(14), 3366–3378. https://doi.org/10.1175/JCLI4189.1
- Hu, Z.-Z., Kumar, A., Huang, B., Zhu, J., & Ren, H. L. (2017). Interdecadal variations of ENSO around 1999/2000. Journal of Applied Genetics, 31(1), 73–81. https://doi.org/10.1007/s13351-017-6074-x
- Hu, Z. Z., Kumar, A., Ren, H. L., Wang, H., L'heureux, M., & Jin, F. F. (2013). Weakened interannual variability in the Tropical Pacific Ocean since 2000. Journal of Climate, 26(8), 2601–2613. https://doi.org/10.1175/jcli-d-12-00265.1

Huang, B., Hu, Z., & Jha, B. (2007). Evolution of model systematic errors in the Tropical Atlantic Basin from coupled climate hindcasts. *Climate Dynamics*, 28, 661–682. https://doi.org/10.1007/s00382-006-0223-8

- Illig, S., & Bachèlery, M. L. (2019). Propagation of subseasonal equatorially-forced coastal trapped waves down to the Benguela Upwelling System. Nature - Scientific Reports, 9, 5306. https://doi.org/10.1038/s41598-019-41847-1
- Illig, S., Bachèlery, M.-L., & Cadier, E. (2018). Subseasonal coastal-trapped wave propagations in the southeastern Pacific and Atlantic Oceans: 2. Wave characteristics and connection with the equatorial variability. *Journal of Geophysical Research: Oceans, 123*, 3942–3961. https://doi.org/10.1029/2017JC013540
- Illig, S., Cadier, E., Bachèlery, M.-L., & Kersalé, M. (2018). Subseasonal coastal-trapped wave propagations in the southeastern Pacific and Atlantic Oceans: 1. A new approach to estimate wave amplitude. *Journal of Geophysical Research: Oceans, 123*, 3915–3941. https://doi. org/10.1029/2017JC013539
- Illig, S., Dewitte, B., Ayoub, N., Du Penhoat, Y., Reverdin, G., De, M., et al. (2004). Interannual long equatorial waves in the tropical Atlantic from a high-resolution ocean general circulation model experiment in 1981–2000. *Journal of Geophysical Research*, 109, C02022. https://doi.org/10.1029/2003JC001771
- Imbol Koungue, R. A., Illig, S., & Rouault, M. (2017). Role of interannual Kelvin wave propagations in the equatorial Atlantic on the Angola Benguela Current system. Journal of Geophysical Research: Oceans, 122, 4685–4703. https://doi.org/10.1002/2016JC012463
- Imbol Koungue, R. A., Rouault, M., Illig, S., Brandt, P., & Jouanno, J. (2019). Benguela Niños and Benguela Niñas in forced ocean simulation from 1958 to 2015. Journal of Geophysical Research: Oceans, 124, 5923–5951. https://doi.org/10.1029/2019JC015013
- Ishii, M., Shouji, A., Sugimoto, S., & Matsumoto, T. (2005). Objective analyses of sea-surface temperature and marine meteorological variables for the 20th century using ICOADS and the Kobe collection. *International Journal of Climatology*, 25(7), 865–879. https://doi. org/10.1002/joc.1169
- Koseki, S., & Imbol Koungue, R. A. (2020). Regional atmospheric response to the Benguela Niñas. International Journal of Climatology, 1–15. https://doi.org/10.1002/joc.6782
- Kumar, A., & Hu, Z.-Z. (2012). Uncertainty in the ocean-atmosphere feedbacks associated with ENSO in the reanalysis products. *Climate Dynamics*, 39(3–4), 575–588. https://doi.org/10.1007/s00382-011-1104-3
- Li, X., Hu, Z., & Becker, E. (2019). On the westward shift of tropical Pacific climate variability since 2000. *Climate Dynamics*, 53, 2905–2918. https://doi.org/10.1007/s00382-019-04666-8

Lübbecke, J. F., Böning, C. W., Keenlyside, N. S., & Xie, S.-P. (2010). On the connection between Benguela and equatorial Atlantic Niños and the role of the South Atlantic Anticyclone. Journal of Geophysical Research, 115, C09015. https://doi.org/10.1029/2009JC005964

- Lübbecke, J. F., Brandt, P., Dengler, M., Kopte, R., Lüdke, J., Richter, I., et al. (2019). Causes and evolution of the southeastern tropical Atlantic warm event in early 2016. *Climate Dynamics*, 53, 261–274. https://doi.org/10.1007/s00382-018-4582-8
- Lübbecke, J. F., Burls, N. J., Reason, C. J., & McPhaden, M. J. (2014). Variability in the South Atlantic anticyclone and the Atlantic Niño mode. Journal of Climate, 27, 8135–8150. https://doi.org/10.1175/JCLI-D-14-00202.1
- Lübbecke, J. F., & McPhaden, M. J. (2014). Assessing the twenty-first-century shift in ENSO variability in terms of the Bjerknes stability index. Journal of Climate, 27(7), 2577–2587. https://doi.org/10.1175/JCLI-D-13-00438.1
- Lutz, K., Rathmann, J., & Jacobeit, J. (2013). Classification of warm and cold water events in the eastern tropical Atlantic Ocean. Atmospheric Science Letters, 14(2), 102–106. https://doi.org/10.1002/asl2.424
- McPhaden, M. J. (2012). A 21st century shift in the relationship between ENSO SST and warm water volume anomalies. *Geophysical Research Letters*, 39, L09706. https://doi.org/10.1029/2012GL051826
- Nnamchi, H., Latif, M., Keenlyside, N., & Park, W. (2020). A boreal summer warming hole in the central Equatorial Atlantic sea surface temperature during the satellite era. *Journal of Geophysical Research: Oceans*, 125, e2019JC015834. https://doi.org/10.1029/ 2019JC015834



- Polo, I., Lazar, A., Rodriguez-Fonseca, B., & Arnault, S. (2008). Oceanic Kelvin waves and tropical Atlantic intraseasonal variability: 1. Kelvin wave characterization. Journal of Geophysical Research, 113, C07009. https://doi.org/10.1029/2007JC004495
- Prigent, A., Lübbecke, J. F., Bayr, T., Latif, M., & Wengel, C. (2020). Weakened SST variability in the tropical Atlantic Ocean since 2000. *Climate Dynamics*, 54, 2731–2744. https://doi.org/10.1007/s00382-020-05138-0
- Rayner, N. A., Parker, D. E., Horton, E. B., Folland, C. K., Alexander, L. V., Rowell, D. P., et al. (2003). Global analyses of sea surface temperature, sea ice, and night marine air temperature since the late nineteenth century. *Journal of Geophysical Research*, 108(D14), 4407. https://doi.org/10.1029/2002JD002670
- Reason, C. J. C., Florenchie, P., Rouault, M., & Veitch, J. (2006). 10 Influences of large scale climate modes and Agulhas System variability on the BCLME region. In V. Shannon, G. Hempel, P. Malanotte-Rizzoli, C. Moloney, & J. Woods (Eds.), *Benguela: Predicting a large marine ecosystem* (Vol. 14, pp. 223–238). Amsterdam: Elsevier. https://doi.org/10.1016/S1570-0461(06)80015-7
- Reynolds, R. W., Smith, T. M., Liu, C., Chelton, D. B., Casey, K. S., & Schlax, M. G. (2007). Daily high-resolution-blended analyses for sea surface temperature. *Journal of Climate*, 20(22), 5473–5496. https://doi.org/10.1175/2007JCLI1824.1
- Richter, I., Behera, S. K., Masumoto, Y., Taguchi, B., Komori, N., & Yamagata, T. (2010). On the triggering of Benguela Niños: Remote equatorial versus local influences. *Geophysical Research Letters*, *37*, L20604. https://doi.org/10.1029/2010GL0444461
- Rouault, M. (2012). Bi-annual intrusion of tropical water in the northern Benguela upwelling. Geophysical Research Letters, 39, L12606. https://doi.org/10.1029/2012GL052099
- Rouault, M., Florenchie, P., Fauchereau, N., & Reason, C. J. C. (2003). South East tropical Atlantic warm events and southern African rainfall. *Geophysical Research Letters*, 30(5), 8009. https://doi.org/10.1029/2002GL014840
- Rouault, M., Illig, S., Bartholomae, C., Reason, C. J. C., & Bentamy, A. (2007). Propagation and origin of warm anomalies in the Angola Benguela upwelling system in 2001. Journal of Marine Systems, 68(3–4), 473–488. https://doi.org/10.1016/j.jmarsys.2006.11.010
- Rouault, M., Illig, S., Lübbecke, J., Imbol, K., & Koungue, R. A. (2018). Origin, development and demise of the 2010–2011 Benguela Niño. Journal of Marine Systems, 188, 39–48. https://doi.org/10.1016/j.jmarsys.2017.07.007
- Saha, S., Moorthi, S., Wu, X., Wang, J., Nadiga, S., Tripp, P., et al. (2014). The NCEP climate forecast system version 2. Journal of Climate, 27, 2185–2208. https://doi.org/10.1175/JCLI-D-12-00823.1
- Servain, J., Caniaux, G., Kouadio, Y. K., McPhaden, M. J., & Araujo, M. (2014). Recent climatic trends in the tropical Atlantic. Climate Dynamics, 43(11), 3071–3089. https://doi.org/10.1007/s00382-014-2168-7
- Shannon, L. V., Boyd, A. J., Brundrit, G. B., & Taunton-Clark, J. (1986). On the existence of an El Niño-type phenomenon in the Benguela system. *Journal of Marine Research*, 44, 495–520. https://doi.org/10.1357/002224086788403105
- Tchipalanga, P., Dengler, M., Brandt, P., Kopte, R., Macuéria, M., Coelho, P., et al. (2018). Eastern boundary circulation and hydrography off Angola: Building Angolan oceanographic capacities. *Bulletin of the American Meteorological Society*, 99, 1589–1605. https://doi.org/ 10.1175/BAMS-D-17-0197.1
- Tokinaga, H., & Xie, S. P. (2011). Weakening of the equatorial Atlantic cold tongue over the past six decades. *Nature Geoscience*, 4(4), 222–226. https://doi.org/10.1038/ngeo1078
- Vizy, E. K., & Cook, K. H. (2016). Understanding long-term (1982–2013) multi-decadal change in the equatorial and subtropical South Atlantic climate. *Climate Dynamics*, 46, 2087–2113. https://doi.org/10.1007/s00382-015-2691-1
- Vizy, E. K., Cook, K. H., & Sun, X. (2018). Decadal change of the South Atlantic Ocean Angola-Benguela frontal zone since 1980. Climate Dynamics, 51, 3251–3273. https://doi.org/10.1007/s00382-018-4077-7